\documentclass[12pt]{article}

\usepackage{fullpage}
\usepackage{nopageno}

\begin{document}

\title{Math 107: Applied Math for Management \\
Spring 2006 \\
Frey 343, TR 8:00-9:30}
\date{}

\maketitle

\begin{itemize}
\item[\bf Instructor] Chris Staecker\\
Email: cstaecker@messiah.edu \\
Phone: x7122\\
Office: Frey 327\\
Office Hours: M 1-2, T 11-12, W 2-3, F 9-10

\item[\bf Course Website] Follow links from {\tt
  http://www.messiah.edu/\~{}cstaecker }

\item[\bf Textbook] Barnett, et al., \emph{Finite Mathematics for business, economics, life sciences, and social sciences}, tenth
edition

\item[\bf Final Exam] Monday 8:00 - 10:00 AM, May 8, 2006

\item[\bf Midterm Exams] Thursday February 23, and Thursday March 30

\item[\bf Homework] will be due every Thursday, and will cover
  material presented in the previous week.

\item[\bf Quizzes] Short (less than 15 minutes) quizzes will be given
  each Friday, covering material presented in the previous week.

\item[\bf Objectives] We will cover basic concepts in finite
  mathematics and their application to business problems. Concepts are
  developed and presented intuitively, without the rigor of technical
  proofs. The course is intended for non-majors.

\item[\bf Grading] The most frequent evaluation will be in the form of
  weekly homework and quizzes. Students will be permitted
to drop the lowest score in each of these categories. Missed quizzes will not
be made up, except in extraordinary circumstances. There
will be two midterm exams and one final exam, which is 
weighted double. The grades are weighed as follows:

\begin{tabular}{lr}
Homework & 35\% \\
Quizzes & 35\% \\
Exams  & 30\%
\end{tabular}

\end{itemize}


\end{document}
